Author's response to reviews

Title: The impact of smoking on adherence to treatment for latent tuberculosis infection

Authors:

Melanie Lavigne (mel_lavigne@hotmail.com)
Isabelle Rocher (isabelle.rocher@inspq.qc.ca)
Colin Steensma (colin.steensma@phac-aspc.gc.ca)
Paul Brassard (paul.brassard@mcgill.ca)

Version: 4 Date: 15 February 2006

Author’s response to reviews: see over
February 15 2006

MS: 2007362780811304
Subject: Formatting changes

Formatting changes requested:

Title and abstract: we made sure the title and abstract are identical in the manuscript and submission system

Author list: We also paid attention to the list of author

Keyword section- We removed the section

Introduction- This was renamed background

Manuscript sections- we followed the order requested thus adding a conclusion and abbreviations section

Section headings-We removed all the unnecessary capitals as requested.

Thank you for your collaboration

Paul Brassard MD